Author's response to reviews

Title: Splenic peliosis with spontaneous splenic rupture - a report of 2 cases.

Authors:

Daniel J. Lashbrook (dan_lashbrook@hotmail.com)
Roger W. James (Roger.James@ruh-bath.swest.nhs.uk)
Andrea J. Phillips (andrea.phillips@ruh-bath.swest.nhs.uk)
Anthony G. Holbrook (Anthony.Holbrook@ruh-bath.swest.nhs.uk)
Andrew C. Agombar (acagombar@yahoo.co.uk)

Version: 6 Date: 25 June 2006

Author's response to reviews:

1) Title page, abstract, competing interests and authors' contributions changed to match format protocol.

2) Section headings changed to sentence case.

3) 'Liver' changed to 'splenic' in Fig 1 legend.

4) 'Fig 2a-c' changed to 'Fig 2'. Figs 2b and 2c deleted.

5) Fig 3a deleted.

6) 'Fig 4a-b' changed to 'Fig 4'. Fig 4b deleted.

7) 'Fig 5a-b' changed to 'Fig 5'. Fig 5b deleted.

8) 'Fig 6a-b' changed to 'Fig 6' and 'Fig 7''

9) 'Fig 6c' changed to 'Fig 8'

10) 'Fig 7a' changed to 'Fig 9'

11) 'Fig 7b-c' changed to 'Fig 10' and 'Fig 11'